Author’s response to reviews

Title: Statins and the risk of type 2 diabetes mellitus: cohort study using the UK Clinical Practice Research Datalink

Authors:

Ana Filipa Macedo (filipa_macedopt@hotmail.com)
Ian Douglas (Ian.Douglas@lshtm.ac.uk)
Liam Smeeth (Liam.Smeeth@lshtm.ac.uk)
Harriet Forbes (Harriet.Forbes@lshtm.ac.uk)
Shah Ebrahim (Shah.Ebrahim@lshtm.ac.uk)

Version: 4 Date: 2 July 2014

Author’s response to reviews:

Manuscript submitted on the 25th March 2014